\documentclass[a4paper,12pt]{article}

\usepackage{anysize}
\usepackage[english]{babel}
%\usepackage[english,spanish]{babel}
%\usepackage[spanish]{babel}
\usepackage{amsmath,amssymb}

\marginsize{2cm}{2cm}{0cm}{2cm}


\begin{document}
\begin{thebibliography}{9}
\bibitem{Chen} X. Chen, Z. C. Gu \& X. G. Wen. arXiv: 1004.3835v2
\bibitem{Lauchli} M. L\"auchli, E. J. Bergholtz \& M. Haque. arXiv: 1003.5656v2 (2010).
\bibitem{Li} H. Li \& F. D. M. Haldane. Phys. Rev. Lett. {\bf{101,}} 010504 (2008).
\bibitem{Sterdyniak1} A. Sterdyniak, A. Chandran, N. Regnault, B. A. Bernevig \& P. Bonderson. arXiv: 1111.2810v2 (2012).
\bibitem{Sterdyniak2} R. Thomale, A. Sterdyniak, N. Regnault \& B. A Bernevig. Phys. Rev. Lett. {\bf{104,}} 180502 (2010).
\bibitem{TSG}  D. C. Tsui, H. L. Stormer, and A. C. Gossard. Two-dimensional magnetotransport in the extreme quantum limit. Phys. Rev. Lett., 48(22):1559�1562, May 1982.
\end{thebibliography}


\title{\bf  Entanglement Spectrum in The Fractional Quantum Hall Effect}

\author{Andr\'es Schlief$^1$ and Alonso Botero$^2$\\
               $^1${\small Universidad de los Andes}\\
               $^2${\small Universidad de los Andes}}
              

\date{22 de Julio de 2012}

\maketitle
\begin{abstract}
\noindent The study of the Fractional Quantum Hall Effect (FQHE) discovered in 1982 by Tsui, St�rmer and Gossard \cite{TSG} has been enlightening in various fields of Physics.
Particullarily, in the past two decades this effect has been of major interest for the study of highly correlated systems entanglement.
During this short communication I will present the concept of \emph{Entanglement Spectrum} and its application to the FQHE \cite{Li}.
In particular, I will present numerical results for Laughlin states in three different geometries
and different Hilbert space partitions \cite{Li,Sterdyniak1, Sterdyniak2}.
A particular emphasis will be given in the interpretation of the entanglement spectra as a clear
sign of the so-called \emph{Topological Order}\cite{Chen}.
In order to do this, the entanglement spectra will be used as numerical evidence of Haldane and Li's conjecture: The low-lying entanglement spectrum corresponds to the energy spectrum of the theory's edge modes described by a particular CFT \cite{Li, Sterdyniak1, Sterdyniak2}. 
\end{abstract}



{\bf Keywords}: Entanglement, Topological Order, Entanglement Spectrum

Modality:  Oral $\; \mid \;$ E-mail: \texttt{af.schlief225@uniandes.edu.co} 


\end{document}

%%% TexDateInsert
%%% Di 9. Apr 05:26:12 CEST 2013
%%% Abstract_Charla_Schlief.tex
